Author's response to reviews

Title: Waist circumference and insulin resistance: a cross-sectional study of Japanese men

Authors:

Shinji Tabata (sthe8916@ba.mbn.or.jp)
Shinichiro Yoshimitsu (yoshimitsu4432shinichiro@yahoo.co.jp)
Tadamichi Hamachi (maxhardt@hotmail.com)
Hiroshi Abe (tttabato@yahoo.co.jp)
Keizo Ohnaka (oonaka@geriat.med.kyushu-u.ac.jp)
Suminori Kono (skono@phealth.med.kyushu-u.ac.jp)

Version: 6 Date: 9 January 2009

Author's response to reviews: see over
January 9, 2009

Re: MS: 4728847332208016 manuscript confirmation required
Title: Waist circumference and insulin resistance: a cross-sectional study of Japanese men

Dear The BioMed Central Editorial Production Team,

Thank you very much for your checking our manuscript. We have finished confirming the content of the above-referenced manuscript. We would like to resubmit a revised version with some corrections. Changes of the content were made at the following points.

1. We corrected the institutional name (line 10 on page 1).

2. We added the word “HOMA-IR” to the explanation of panel C for the definition of insulin resistance (line 8 on Page 16; Figure legends).

3. We corrected the spelling mistake of “serum” (line 11 on Page 17; Table 1)

We should appreciate your consideration once again.

Best wishes,

Shinji Tabata, MD, PhD
Corresponding author